Author’s response to reviews

Title: Diabetes insipidus and panhypopituitarism as a first presentation of silent adenocarcinoma of lung: a case report and literature review

Authors:

Sirinart Sirinvaravong (sirinart.sir@mahidol.ac.th;sirinart.oh@gmail.com)
Peeradon Vibhatavata (p_wipatavut@hotmail.com)
Pawee Chumpharorit (pawee@yahoo.com)
Pornsuk Cheunsuchon (pcheunsuchon@gmail.com)
Sutin Sriussadaporn (sutin.sri@mahidol.ac.th)

Version: 3 Date: 09 Oct 2019

Author’s response to reviews:

BMC Endocrine Disorders

9 October 2019

Dear the editor and reviewers,

Reference number: BEND-D-19-00184

Thank you for your letter and comments concerning our manuscript entitled “Diabetes insipidus and panhypopituitarism as a first presentation of silent adenocarcinoma of lung: a case report and literature review: Reference number: BEND-D-19-00184”. We studied your comments carefully and made corrections in the manuscript which we hope to meet with your approval. We answered your comments in details in the following texts.

Replies to the editor comments

Editor Comments: Please add the legend of figure 5.
Reply: Legend of figure 5 was added into the manuscript (section “figure legend”, page 26, line 351-354)

“Figure 5. Tumor resected from sellar mass composing of malignant epithelial cells with glandular formation (A, hematoxylin-eosin, x100) and intracellular mucin (B, x200). Immunohistochemical study showed that tumor cells were immunoreactive for antibodies of cytokeratin 7 (C, x200) and TTF-1 (E, x200) but not cytokeratin 20 (D, x200) and CDX-2 (F, x200).”

We really hope this modification meet with your approval.

We have submitted a clean version of the manuscript. All track changes and highlights were removed.

Yours Sincerely,

Sirinart Sirinvaravong, MD

Division of Endocrinology and Metabolism, Department of Medicine

Faculty of Medicine Siriraj Hospital, Mahidol University, Bangkok 10700, Thailand

Email: Sirinart.sir@mahidol.ac.th